Author's response to reviews

Title: A Qualitative Exploration of Malaysian Cancer Patients' Perspectives on Cancer and its treatment

Authors:

Maryam Farooqui (maryamfaroogui.uitm@gmail.com)
Mohamed Azmi Hassali (azmihassali@gmail.com)
Aishah Knight Shatar (Aknight@amdi.usm.edu.my)
Asrul Akmal Shafie (aakmal@usm.my)
Tan Boon Seang (bseang@yahoo.com.sg)
Muhammad Aslam Farooqui (farooqui@acms.edu.my)

Version: 4 Date: 20 June 2011

Author's response to reviews: see over
Thank you very much for your comments on my manuscript MS: 548127975157456. Below are my responses to the requests made by the editorial board:

**Comments:** Competing interests: Please include a 'Competing interests' section after the Conclusions. If there are none to declare, please write 'The authors declare that they have no competing interests'. Please check the instructions for authors on the journal website for a full list of questions to consider when writing your competing interests statement.

**Response:** Modified as suggested

**Comments:** Authors' information: Please place the Authors' Contributions section after Competing interests. Please check the instructions for authors on the journal website for the correct format to use for Authors' Contributions.

**Response:** Modified as suggested

**Comments:** Tables: Please ensure that the order in which your tables are cited is the same as the order in which they are provided. Every table must be cited in the text, using Arabic numerals. Please do not use ranges when listing tables. Tables must not be subdivided, or contain tables within tables. Please note that we are unable to display vertical lines or text within tables, no display merged cells: please re-layout your table without these elements. Tables should be formatted using the Table tool in your word processor. Please ensure the table title is above the table and the legend is below the table. For more information, see the instructions for authors on the journal website.

**Response:** Modified as suggested

**Comments:** Questionnaire: please include a copy of your questionnaire as an Additional File, properly cited in the Methods section.

**Response:** Modified as suggested

Thank you very much, I look forward for your kind response

Maryam Farooqui
Faculty of Pharmacy
Universiti Teknologi MARA (UiTM), Pulau Pinang
13500, Seberang Perai, Pulau Pinang
Malaysia